Author's response to reviews

Title: Clinical, socio-demographic and radiological predictors of short-term outcome in rotator cuff disease

Authors:

Ole M Ekeberg (o.m.ekeberg@medisin.uio.no)
Erik Bautz-Holter (erik.bautz-holter@medisin.uio.no)
Niels G Juel (Niels.Gunnar.Juel@oslo-universitetssykehus.no)
kaia B Engebretsen (k.b.engebretsen@medisin.uio.no)
Synnøve Kvalheim (synnove.kvalheim@oslo-universitetssykehus.no)
Jens I Brox (jens.ivar.brox@oslo-universitetssykehus.no)

Version: 5 Date: 11 October 2010

Author's response to reviews: see over
Dear BMC Series Editorial Production Team

Enclosed is a revised version of the article entitled “Clinical, socio-demographic and radiological predictors of short-term outcome in rotator cuff disease”.

Following is a point-by-point response to the concerns of the editors:

We have done a careful proofreading of the manuscript.

Major revisions:
- The list of authors in the manuscript is now written exactly as they are in the submission system, both in style and order.
- The tables are copied into the manuscript.

Minor revisions:
- Line numbers are removed
- Highlighting and tracking are removed from the manuscript

Please contact me if you have any questions.

Best regards

Ole M. Ekeberg, MD, PhD student
Department of Physical Medicine and Rehabilitation
Ulleval University Hospital, N-0407 Oslo, Norway.

tel: + 47 23027444
fax: + 47 23027455
e-mail: ole.m.ekeberg@medisin.uio.no